Author's response to reviews

Title: The future prospects of Lithuanian family physicians: a 10-year forecasting study

Authors:

Liudvika Starkiene (liudvika.starkiene@sam.lt)
Kastytis Smigelskas (kastsmig@yahoo.com)
Zilvinas Padaiga (padaiga@kmu.lt)
Jack Reamy (reamy@xavier.edu)

Version: 5 Date: 26 September 2005

Author's response to reviews:

Dear Editors,
We are ready to submit the final version of the manuscript as we have applied the following requested formatting changes:
Title & Abstract - these are identical in the manuscript and in the submission system.
Author list - The list of authors in the manuscript is written exactly as they are in the submission system. The authors (forename, initial (if any), surname) are written identically and in the order in which they are listed in the submission system.
Competing interests - We have changed the statement "None to declare" to 'The authors declare that they have no competing interests'.
Figures - The figure was cropped as closely as possible to minimise white space around the image according to the online figure cropping guide.

On behalf of the authors,
Liudvika Starkiene